Author's response to reviews

Title: Comprehensive analyses of imprinted differentially methylated regions reveal epigenetic and genetic characteristics in hepatoblastoma.

Authors:

Janette M Rumbajan (janette_mr@ymail.com)
Toshiyuki Maeda (topgun.mae@gmail.com)
Ryota Souzaki (ryotas@pedsurg.med.kyushu-u.ac.jp)
Kazumasa Mitsui (k-mitsui@med.toho-u.ac.jp)
Ken Higashimoto (higashim@cc.saga-u.ac.jp)
Kazuhiko Nakabayashi (knakabayashi@nch.go.jp)
Hitomi Yatsuki (yatsuki@cc.saga-u.ac.jp)
Kenichi Nishioka (nshoka@cc.saga-u.ac.jp)
Ryoko Harada (r.harada@med.toho-u.ac.jp)
Kenichi Kohashi (kohas@surgpath.med.kyushu-u.ac.jp)
Yoshinao Oda (oda@surgpath.med.kyushu-u.ac.jp)
Kenichiro Hata (khata@nch.go.jp)
Tsutomu Saji (saji34ben@med.toho-u.ac.jp)
Tomoaki Taguchi (taguchi@pedsurg.med.kyushu-u.ac.jp)
Tatsuro Tajiri (taji@koto.kpu-m.ac.jp)
Hidenobu Soejima (soejimah@cc.saga-u.ac.jp)
Keiichiro Joh (joh@cc.saga-u.ac.jp)

Version: 3 Date: 4 April 2013

Author's response to reviews:

We have made a minor revision in "Samples" of Methods section. A following sentence was added.

Written informed consents were obtained from the parents or the guardians of the participants, because the participants were children or dead.